\begin{thebibliography}{1}

\bibitem{AlKaWo:97}
A.~ALFAKIH, A.~KHANDANI, and H.~WOLKOWICZ.
\newblock Solving {E}uclidean distance matrix completion problems via
  semidefinite programming.
\newblock {\em Comput. Optim. Appl.}, 12(1-3):13--30, 1999.
\newblock Computational optimization---a tribute to Olvi Mangasarian, Part I.

\bibitem{AlWo:98}
A.~ALFAKIH and H.~WOLKOWICZ.
\newblock On the embeddability of weighted graphs in {E}uclidean spaces.
\newblock Technical Report CORR 98-12, University of Waterloo, 1998.

\bibitem{AlWo:99}
A.~ALFAKIH and H.~WOLKOWICZ.
\newblock Matrix completion problems.
\newblock In H.~Wolkowicz, R.~Saigal, and L.~Vandenberghe, editors, {\em
  HANDBOOK OF SEMIDEFINITE PROGRAMMING: Theory, Algorithms, and Applications},
  page~11. Kluwer Academic Publishers, Boston, MA, 2000.

\bibitem{GrJoSaWo:84}
B.~GRONE, C.R. JOHNSON, E.~MARQUES~de SA, and H.~WOLKOWICZ.
\newblock Positive definite completions of partial {H}ermitian matrices.
\newblock {\em Linear Algebra Appl.}, 58:109--124, 1984.

\bibitem{HeReVaWo:93}
C.~HELMBERG, F.~RENDL, R.~J. VANDERBEI, and H.~WOLKOWICZ.
\newblock An interior-point method for semidefinite programming.
\newblock {\em SIAM J. Optim.}, 6(2):342--361, 1996.

\bibitem{JoKrWo:95}
C.R. JOHNSON, B.~KROSCHEL, and H.~WOLKOWICZ.
\newblock An interior-point method for approximate positive semidefinite
  completions.
\newblock {\em Comput. Optim. Appl.}, 9(2):175--190, 1998.

\end{thebibliography}

\documentclass{article}
\usepackage{latexsym}
\input epsf
\newtheorem{exam}{Example}
\newtheorem{prop}{Proposition}
\newtheorem{lem}{Lemma}
\newtheorem{thm}{Theorem}
\newtheorem{cor}{Corollary}
\newtheorem{alg}{Algorithm}
\newcommand{\sn}{{\cal S}^n }
\newcommand{\Sn}{{\cal S}^n }
\newcommand{\hn}{{\cal H}^n }
\newcommand{\p}{{\cal P} }
\newcommand{\g}{{\cal G} }
\newcommand{\kvec}{{\rm vec\,}}
\newcommand{\adj}{{\rm adj\,}}
\newcommand{\trace}{{\rm trace\,}}
\newcommand{\tr}{{\rm trace\,}}
\newcommand{\diag}{{\rm diag\,}}
\newcommand{\Diag}{{\rm Diag\,}}
\newcommand{\Se}{{\mathcal S}_e }
\newcommand{\Sd}{{\mathcal S}_d }
\newcommand{\Sc}{{\mathcal S}_C }
\newcommand{\Sh}{{\mathcal S}_H }
\newcommand{\snn}{{\mathcal S}_{n-1} }
\newcommand{\GG}{{\mathcal G} }
\newcommand{\KK}{{\mathcal K} }
\newcommand{\DD}{{\mathcal D} }
\newcommand{\BB}{{\mathcal B} }
\newcommand{\PP}{{\mathcal P} }
\newcommand{\TT}{{\mathcal T} }
\newcommand\T{{\mathcal T}}
\newcommand{\bt}{ \begin{tabular} }
\newcommand{\et}{ \end{tabular} }
\newcommand\A{{\mathcal A}}
\newcommand\E{{\mathcal E}}
\newcommand{\bpr}{{\bf Proof.} \hspace{1 em}}
\newcommand{\beq}{\begin{equation}}
\newcommand{\eeq}{\end{equation}}
%\newcommand{\epr}{\\ \hspace*{4.5in}  $\Box$}
\newcommand{\QED}{\hfill ~\rule[-1pt] {8pt}{8pt}\par\medskip ~~}
\newcommand{\epr}{\QED}
\begin{document}
\bibliographystyle{plain}

\nocite{GrJoSaWo:84,HeReVaWo:93,JoKrWo:95,AlWo:99,AlWo:98,AlKaWo:97}
\bibliography{.psd,.master,.publs}
\end{document}
